PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	The impact of COVID-19 on the the digital divide: A rapid review
AUTHORS	Litchfield, Ian; Shukla, David; Greenfield, Sheila

VERSION 1 – REVIEW

REVIEWER	Nuno Martins IPCA and ID+, School of Design
REVIEW RETURNED	27-May-2021

CENEDAL COMMENTS	The study presented in this paper is highly relevant. The adaption of
GENERAL COMMENTS	The study presented in this paper is highly relevant. The adoption of
	digital communication tools in the Health area is a subject that
	generates much discussion and is highly complex. The tendency is
	the growing recognition of its importance. However, this paper
	reinforces the identified problem of asymmetries among the
	population. Asymmetries are high, mainly affecting the most
	vulnerable groups in social, cultural, economic and literacy terms.
	The analysis presented in this paper reinforces that there is still a lot
	to be done to make digital media effectively relevant to society at
	large. To evaluate the importance of digital media only through a set
	of population niches is to have a distorted view of reality. And this
	paper makes an important contribution to that clarification.
	The paper is well structured, with a very clear presentation of the
	methodologies and the process developed, as well as its limitations.
	The only recommendation is that the conclusions chapter should be
	a little more developed.

REVIEWER Sonu Bhaskar South Western Sydney Local Health District, Liverpool Hospital, Department of Neurology & Neurophysiology REVIEW RETURNED 07-Jun-2021
Department of Neurology & Neurophysiology
REVIEW RETURNED 07-Jun-2021
GENERAL COMMENTS This study on the digital divide during the first wave of the pandem from Litchfield et al is a well-conceived, timely and well-written pier The authors highlight the systemic disparities in elderly and ethnic minorities. Thank you for your invitation to review this work. I have a few suggestions/comments that I would invite authors to reflect upon; 1. In the abstract/conclusion, please consider adding "During the early phase of COVID-19 pandemic, in the developed world, famil difficulties in utilisation" Since this review was focussed on the developed world and on the early phase of the pandemic. 2. Introduction (Page 4/Lines 32-39): Not just communities, discrepancies exist across geographics and specialities. Suggest adding briefly this to the statement. (See https://pubmed.ncbi.nlm.nih.gov/33178656/, https://pubmed.ncbi.nlm.nih.gov/33014958/) 3. Page 4, Lines 56-59, please revise the sentence and provide a

appropriate reference.
4. Minor change: Page 5: Lines 57-59: Please remove space after
"(PRISMA)" and add space before "to"
5. Page 6: Inclusion criteria: Lines 39-42: After "within the developed
world, please add ", during the early phase of the pandemic, "
6. Authors have expanded on the framework initially developed by
Ai-Chi Loh et al, therefore suggest removing "developed a novel
framework" from the manuscript.
7. Page 9: Lines 21-23: Please specify which vulnerable patients?
8. In the Methods, it is suggested to include a definition of what
defined "Developed world" - according to world bank definition etc?
9. Page 13: Lines 41-43: Please cite the two studies discussed in
the first statement.
10. Page 15-16: Strengths and Weakness: Please add further
discussion on that this study only analysed studies during the early
phase of the pandemic. Secondly, no quantitative synthesis was
performed - consistent with the limitations of narrative/rapid review.
Also, only limited studies were of high quality and quality
(methodological and risk of bias) assessment of individual studies were not performed.
11. Page 17: Lines 49-54 suggest also adding other relevant calls:
https://pubmed.ncbi.nlm.nih.gov/33810572/,
https://pubmed.ncbi.nlm.nih.gov/33343410/)
12. Page 18: Lines 43-45: In addition to face-time technologies,
other platforms such as online blogs for psychiatric patients have
also been deployed and shown benefit in the vulnerable populations
(see the work by Lehner et al from Germany
https://pubmed.ncbi.nlm.nih.gov/33425836/). Suggest adding a brief
discussion on this.
13. In the Conclusion, please specify that these findings are to be
interpreted in the context of the developed world.

REVIEWER	K Cooper Robert Gordon University, School of Health Sciences
REVIEW RETURNED	07-Jun-2021

GENERAL COMMENTS	Thank you for the opportunity to review this topical and interesting rapid review.
	•
	This is a really important subject area and pulling together the
	research to date will be useful for a number of stakeholders.
	I particularly liked the use of the Framework for interpreting the digital divide.
	I think a few areas of the manuscript could be enhanced:
	1. Page 4, line 50 - not clear what "this" is being referred to
	2. As there is no single agreed methodology for a rapid review, there
	needs to be a short section that describes & justifies what methods
	were used and what made this a rapid review (e.g. the limited
	number of databases searched?) other than it being done quickly
	3. For transparency and for readers to assess the likelihood of
	papers being missed, I would recommend a full, detailed search
	strategty for each of the 3 databases searched presented as an
	Appendix
	4. Page 15/16 strengths & weaknesses - this section needs further
	consideration of the limitatrions of a rapid review - i.e. limited
	databases searched, no appraisal of methodological quality etc &
	what this means for recommendations that can be made from the
	review
	Minor comments
	Page 4, line 56 - word missing in sentence
	Needs consistent use of COVID-19 throughout manuscipt

Page 19, line 12 - don;t think SES has been previously defined - also done with ED earlier in manuscipt
--

VERSION 1 – AUTHOR RESPONSE

Reviewer: 1

Dr. Nuno Martins, IPCA and ID+

Comments to the Author:

The study presented in this paper is highly relevant. The adoption of digital communication tools in the Health area is a subject that generates much discussion and is highly complex. The tendency is the growing recognition of its importance. However, this paper reinforces the identified problem of asymmetries among the population. Asymmetries are high, mainly affecting the most vulnerable groups in social, cultural, economic and literacy terms.

The analysis presented in this paper reinforces that there is still a lot to be done to make digital media effectively relevant to society at large. To evaluate the importance of digital media only through a set of population niches is to have a distorted view of reality. And this paper makes an important contribution to that clarification.

The paper is well structured, with a very clear presentation of the methodologies and the process developed, as well as its limitations.

Thank you

The only recommendation is that the conclusions chapter should be a little more developed.

Reviewer: 2

Dr. Sonu Bhaskar, South Western Sydney Local Health District, Ingham Institute Comments to the Author:

This study on the digital divide during the first wave of the pandemic from Litchfield et al is a wellconceived, timely and well-written piece. The authors highlight the systemic disparities in elderly and ethnic minorities.

Thank you

Thank you for your invitation to review this work. I have a few suggestions/comments that I would invite authors to reflect upon;

1. In the abstract/conclusion, please consider adding "During the early phase of COVID-19 pandemic, in the developed world, familiar difficulties in utilisation..." Since this review was focussed on the developed world and on the early phase of the pandemic.

Thank you for the suggestion, following your advice the text now reads,

"During the early phase of the COVID-19 pandemic, in the developed world, familiar difficulties..."

2. Introduction (Page 4/Lines 32-39): Not just communities, discrepancies exist across geographics and specialities. Suggest adding briefly this to the statement. (See https://pubmed.ncbi.nlm.nih.gov/33178656/,

https://pubmed.ncbi.nlm.nih.gov/33014958/)

Thank you for bringing these important publications to our attention. The text now reads,

"However persistent discrepancies exist across geographies and between communities in how they access and utilise digital technologies, differences compounded by the growing sophistication in the functionality of devices and connectivity (16-18)."

3. Page 4, Lines 56-59, please revise the sentence and provide an appropriate reference.

Thank you. The sentence has now been revised and reads,

"Despite acknowledged inequities in digital access and utilisation, measures introduced to reduce infection rates following the onset of the COVID- 19 pandemic in spring 2020, led to an acceleration of the reliance on digital health technologies both in Europe and the United States of America (19-22)"

4. Minor change: Page 5: Lines 57-59: Please remove space after "(PRISMA)" and add space before "to"

Thank you, now amended as advised.

5. Page 6: Inclusion criteria: Lines 39-42: After "within the developed world, please add "..., during the early phase of the pandemic, "

Thank you. This has now been clarified in the text which reads,

"The focus of our work was the provision of care within the developed world during the early phase of the pandemic to ensure relevance for policymakers, commissioners..."

6. Authors have expanded on the framework initially developed by Ai-Chi Loh et al, therefore suggest removing "developed a novel framework" from the manuscript.

Thank you. We have now changed the wording throughout to reflect your advice and referred to the framework as being "informed by ..." or "based on..."

7. Page 9: Lines 21-23: Please specify which vulnerable patients?

We are referring to vulnerable as the phrase defined by Hughes et al and classified by National Health Service England (23). The text now reads,

"One study looked at Digital Access; set in UK primary care it explored internet connectivity amongst vulnerable patients (including those that have received an organ transplant, are undertaking immunotherapy or a course of radiotherapy for lung cancer) (11, 23)."

8. In the Methods, it is suggested to include a definition of what defined "Developed world" - according to World Bank definition etc?

Thank you for the notification, we have now include the text,

"The focus of our work was the provision of care within the developed world (i.e. one which is predominantly industrialized and more economically developed with a higher individual income) (24)"

9. Page 13: Lines 41-43: Please cite the two studies discussed in the first statement.

Thank you, the text now reads...

"Two studies used routinely collected electronic health data, though were conducted independently of

each other at two different sites in New York (United States) (25, 26). The first used data gathered

from patients at New York University Hospital collected over a 6-week period to determine whether

they had received their COVID-19 diagnosis at an office visit or via video consultation."

10. Page 15-16: Strengths and Weakness: Please add further discussion on that this study only analysed studies during the early phase of the pandemic.

Thank you, we have now included the text.

"We focussed on the early phase of the pandemic to understand the impact of the rapid changes to service delivery on those most vulnerable to the digital divide with the intention of producing timely findings that might inform service delivery in subsequent phases."

Secondly, no quantitative synthesis was performed - consistent with the limitations of narrative/rapid review.

Thank you we have now included the text,

"Our rapid review discovered only a small numbers of heterogeneous papers of limited geographic scope which precluded data synthesis and may have introduced a degree of bias."

Also, only limited studies were of high quality and quality (methodological and risk of bias) assessment of individual studies were not performed.

Thank you we have now acknowledged this in the text which reads,

"We did not formally assess the quality of the papers to facilitate the timeliness of the review process as with previous rapid reviews (14)"

11. Page 17: Lines 49-54 suggest also adding other relevant calls: https://pubmed.ncbi.nlm.nih.gov/33810572/, https://pubmed.ncbi.nlm.nih.gov/33343410/)

Thank you, these have now been cited.

12. Page 18: Lines 43-45: In addition to face-time technologies, other platforms such as online blogs for psychiatric patients have also been deployed and shown benefit in the vulnerable populations (see the work by Lehner et al from Germany <u>https://pubmed.ncbi.nlm.nih.gov/33425836/</u>). Suggest adding a brief discussion on this.

Thank you, we have now included the line,

"...(27) and related work found benefits from the introduction of an online blog tailored for psychiatric patients (28)."

13. In the Conclusion, please specify that these findings are to be interpreted in the context of the developed world.

Thank you, now clarified in the text which reads,

"The rapid incorporation of digital technologies into mainstream health care delivery due to the COVID pandemic was widely understood and accepted by patients in the developed world unwilling to breach social distancing advice."

Reviewer: 3 Dr. K Cooper, Robert Gordon University Comments to the Author: Thank you for the opportunity to review this topical and interesting rapid review. This is a really important subject area and pulling together the research to date will be useful for a number of stakeholders. I particularly liked the use of the Framework for interpreting the digital divide.

Thank you

I think a few areas of the manuscript could be enhanced: 1. Page 4, line 50 - not clear what "this" is being referred to

Thank you, now clarified in the text,

"This rapid introduction led to concerns that the new digitally reliant models of healthcare delivery will disproportionately affect the health of disadvantaged communities (5, 29-31) such as ethnic minorities (32), rural populations (4), the elderly (33) and residents of care homes (34)."

2. As there is no single agreed methodology for a rapid review, there needs to be a short section that describes & justifies what methods were used and what made this a rapid review (e.g. the limited number of databases searched?) other than it being done quickly

Thank you for advice. We have now clarified this in the text under study design,

"Rapid reviews have previously been recommended by the WHO amongst others for their ability to provide timely and credible evidence for policymakers and practitioners in what is a dynamic and evolving public health crisis (35, 36). We have used many of the principles of a systematic review process; our search terms were clearly defined using Boolean principles and the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) described the search (37). The inclusion and exclusion criteria were clearly defined and two reviewers agreed on the selection of the various papers (The search terms can be found in supplementary file 1). However the systematic review process was expedited by amending several steps i.e. drawing only on the major medical databases and forgoing a structured appraisal of the quality of selected studies in place of a transparent description of the characteristics of each within the results."

3. For transparency and for readers to assess the likelihood of papers being missed, I would recommend a full, detailed search strategy for each of the 3 databases searched presented as an Appendix

Thank you now included as a supplementary file.

4. Page 15/16 strengths & weaknesses - this section needs further consideration of the limitations of a rapid review - i.e. limited databases searched, no appraisal of methodological quality etc & what this means for recommendations that can be made from the review

Thank you, we have now discussed this in the Strengths & weaknesses section and the text now reads,

"Our rapid review discovered only a small numbers of heterogeneous papers of limited geographic scope which precluded data synthesis and may have introduced a degree of bias. The lack of a theoretical underpinning in many of the papers limited generalisability (10) and that two of the studies relied on self-reported data (11, 12) raises familiar issues regards their reliability (13). However previous comparisons between systematic and rapid reviews have failed to find significant differences in the outcomes they report (14, 15) and all of our included studies offered valuable insight into how the digital divide was magnified by the changes to health delivery in the early stages of the pandemic."

Minor comments Page 4, line 56 - word missing in sentence

Thank you, line now reads

"Despite the acknowledged inequities in digital access and utilisation, measures introduced to reduce infection rates following the onset of the COVID- 19 pandemic..."

Needs consistent use of COVID-19 throughout manuscript

Thank you, now amended throughout.

Page 10 - line 28 - "ethical" should be ethnic I presume

Yes it should - thank you, now amended.

Page 19, line 12 – don't think SES has been previously defined - also done with ED earlier in manuscript

Thank you now amended.

Reference List

1. England N. Standard operating procedure for for general practice in the context of coronavirus (COVID-19) 2020 [Available from:

https://www.england.nhs.uk/coronavirus/publication/managing-coronavirus-covid-19-in-general-practice-sop/.

2. Facher L. 9 ways Covid-19 may forever upend the U.S. health care industry 2020 [Available from: https://www.statnews.com/2020/05/19/9-ways-covid-19-forever-upend-health-care/.

3. D2N2. New European Social Fund call to help address the digital divide. 2020.

4. Lee JGL, LePrevost CE, Harwell EL, Bloss JE, Cofie LE, Wiggins MF, et al. Coronavirus pandemic highlights critical gaps in rural Internet access for migrant and seasonal farmworkers: a call for partnership with medical libraries. J Med Libr Assoc. 2020;108(4):651-5.

Watts G. COVID-19 and the digital divide in the UK. Lancet Digit Health. 2020;2(8):e395-e6.
 Allman K. 2020. [cited 2020].

7. Majid A. 2020. Available from: <u>https://blogs.bmj.com/bmj/2020/09/01/covid-19-is-magnifying-the-digital-divide/</u>.

8. Stanford University. Digital Divide 2020 [Available from:

https://cs.stanford.edu/people/eroberts/cs181/projects/digital-divide/start.html

9. Medicine JHUo. Mortality Analyses. 2021.

10. Davidoff F, Dixon-Woods M, Leviton L, Michie S. Demystifying theory and its use in improvement. BMJ Quality & amp; app; Safety. 2015;24(3):228.

11. Hughes T, Beard E, Bowman A, Chan J, Gadsby K, Hughes M, et al. Medical student support for vulnerable patients during COVID-19 - a convergent mixed-methods study. BMC Med Educ. 2020;20(1):377.

12. Campos-Castillo C, Anthony D. Racial and ethnic differences in self-reported telehealth use during the COVID-19 pandemic: a secondary analysis of a US survey of internet users from late March. Journal of the American Medical Informatics Association. 2020.

13. Jerolmack C, Khan S. Talk Is Cheap: Ethnography and the Attitudinal Fallacy. Sociological Methods & Research. 2014;43(2):178-209.

14. Tricco AC, Antony J, Zarin W, Strifler L, Ghassemi M, Ivory J, et al. A scoping review of rapid review methods. BMC Medicine. 2015;13(1):224.

15. Abou-Setta AM, Jeyaraman M, Attia A, Al-Inany HG, Ferri M, Ansari MT, et al. Methods for Developing Evidence Reviews in Short Periods of Time: A Scoping Review. PloS one. 2016;11(12):e0165903-e.

16. van Deursen AJAM, van Dijk JAGM. The first-level digital divide shifts from inequalities in physical access to inequalities in material access. New Media & Society. 2018;21(2):354-75.

17. Bhaskar S, Bradley S, Chattu VK, Adisesh A, Nurtazina A, Kyrykbayeva S, et al. Telemedicine Across the Globe-Position Paper From the COVID-19 Pandemic Health System Resilience PROGRAM (REPROGRAM) International Consortium (Part 1). Frontiers in public health. 2020;8:556720-.

18. Bhaskar S, Bradley S, Chattu VK, Adisesh A, Nurtazina A, Kyrykbayeva S, et al. Telemedicine as the New Outpatient Clinic Gone Digital: Position Paper From the Pandemic Health System REsilience PROGRAM (REPROGRAM) International Consortium (Part 2). Front Public Health. 2020;8:410.

19. Reeves JJ, Ayers JW, Longhurst CA. Telehealth in the COVID-19 Era: A Balancing Act to Avoid Harm. Journal of medical Internet research. 2021;23(2):e24785-e.

20. Evans C. The coronavirus crisis and the technology sector. Business Economics. 2020;55(4):253-66.

21. Thornton J. Covid-19: how coronavirus will change the face of general practice forever. BMJ. 2020;368:m1279.

Webster P. Virtual health care in the era of COVID-19. Lancet. 2020;395(10231):1180-1.
NHSE. Who is at high risk from coronavirus (clinically extremely vulnerable) 2020 [Available

from: <u>https://www.nhs.uk/conditions/coronavirus-covid-19/people-at-higher-risk/who-is-at-high-risk-from-coronavirus-clinically-extremely-vulnerable/</u>.

24. Paprotny D. Convergence between Developed and Developing countries: A centennial perspective. Social indicators research. 2021;153(1):193-225.

25. Weber E, Miller SJ, Astha V, Janevic T, Benn E. Characteristics of Telehealth Users in NYC for COVID-related Care during the Coronavirus Pandemic. J Am Med Inform Assoc. 2020.

26. Chunara R, Zhao Y, Chen J, Lawrence K, Testa PA, Nov O, et al. Telemedicine and healthcare disparities: a cohort study in a large healthcare system in New York City during COVID-19. Journal of the American Medical Informatics Association. 2020.

27. Runfola M, Fantola G, Pintus S, Iafrancesco M, Moroni R. Telemedicine Implementation on a Bariatric Outpatient Clinic During COVID-19 Pandemic in Italy: an Unexpected Hill-Start. Obes Surg. 2020;30(12):5145-9.

28. Lehner A, Nuißl K, Schlee W, Langguth B. Staying Connected: Reaching Out to Psychiatric Patients During the Covid-19 Lockdown Using an Online Blog. Front Public Health. 2020;8:592618.

29. The Good Things Foundation. Digital Inclusion in Health and Care: Lessons learned from the NHS Widening Digital Participation Programme (2017-2020). 2020.

30. Ramsetty A, Adams C. Impact of the digital divide in the age of COVID-19. Journal of the American Medical Informatics Association. 2020;27(7):1147-8.

31. De R, Pandey N, Pal A. Impact of digital surge during Covid-19 pandemic: A viewpoint on research and practice. Int J Inf Manage. 2020;55:102171-.

32. Scanzera AC, Kim SJ, Paul Chan RV. Teleophthalmology and the digital divide: inequities highlighted by the COVID-19 pandemic. Eye. 2020.

33. Martins Van Jaarsveld G. The Effects of COVID-19 Among the Elderly Population: A Case for Closing the Digital Divide. Frontiers in Psychiatry. 2020;11(1211).

34. Seifert A, Batsis JA, Smith AC. Telemedicine in Long-Term Care Facilities During and Beyond COVID-19: Challenges Caused by the Digital Divide. Frontiers in Public Health. 2020;8(690).

 Oliver K, Innvar S, Lorenc T, Woodman J, Thomas J. A systematic review of barriers to and facilitators of the use of evidence by policymakers. BMC health services research. 2014;14(1):1-12.
 Langlois EV, Straus SE, Antony J, King VJ, Tricco AC. Using rapid reviews to strengthen health policy and systems and progress towards universal health coverage. BMJ Global Health. 2019;4(1):e001178.

37. Moher D, Shamseer L, Clarke M, Ghersi D, Liberati A, Petticrew M, et al. Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015 statement. Syst Rev. 2015;4(1):1-.